



#### https://helda.helsinki.fi

## Genetic analyses implicate complex links between adult testosterone levels and health and disease

### FinnGen Consortium

2023-01-18

FinnGen Consortium , Leinonen , J T , Mars , N , Lehtonen , L E , Ahola-Olli , A , Ruotsalainen , S , Lehtimäki , T , Kähönen , M , Raitakari , O , Piltonen , T , Tuomi , T , Daly , M , Ripatti , S , Tukiainen , T & Pirinen , M 2023 , ' Genetic analyses implicate complex links between adult testosterone levels and health and disease ' , Communications Medicine , vol. 3 , 4 . https://doi.org/10.1038/s43856-022-00226-0

http://hdl.handle.net/10138/354606 https://doi.org/10.1038/s43856-022-00226-0

cc\_by publishedVersion

Downloaded from Helda, University of Helsinki institutional repository.

This is an electronic reprint of the original article.

This reprint may differ from the original in pagination and typographic detail.

Please cite the original version.

## communications medicine

#### ARTICLE

https://doi.org/10.1038/s43856-022-00226-0

#### Check for updates

# Genetic analyses implicate complex links between adult testosterone levels and health and disease

**OPEN** 

Jaakko T. Leinonen<sup>® 1</sup>, Nina Mars<sup>® 1,26</sup>, Leevi E. Lehtonen<sup>1,26</sup>, Ari Ahola-Olli<sup>1,26</sup>, Sanni Ruotsalainen<sup>® 1</sup>, Terho Lehtimäki<sup>® 2,3,4</sup>, Mika Kähönen<sup>3,5</sup>, Olli Raitakari<sup>6,7,8</sup>, FinnGen Consortium\*, Terhi Piltonen<sup>® 9</sup>, Mark Daly<sup>1,10</sup>, Tiinamaija Tuomi<sup>1,11,12,13,14</sup>, Samuli Ripatti<sup>® 1,10,15</sup>, Matti Pirinen<sup>® 1,15,16</sup> & Taru Tukiainen<sup>® 1⊠</sup>

#### Abstract

**Background** Testosterone levels are linked with diverse characteristics of human health, yet, whether these associations reflect correlation or causation remains debated. Here, we provide a broad perspective on the role of genetically determined testosterone on complex diseases in both sexes.

**Methods** Leveraging genetic and health registry data from the UK Biobank and FinnGen (total N = 625,650), we constructed polygenic scores (PGS) for total testosterone, sexhormone binding globulin (SHBG) and free testosterone, associating these with 36 endpoints across different disease categories in the FinnGen. These analyses were combined with Mendelian Randomization (MR) and cross-sex PGS analyses to address causality.

**Results** We show testosterone and SHBG levels are intricately tied to metabolic health, but report lack of causality behind most associations, including type 2 diabetes (T2D). Across other disease domains, including 13 behavioral and neurological diseases, we similarly find little evidence for a substantial contribution from normal variation in testosterone levels. We nonetheless find genetically predicted testosterone affects many sex-specific traits, with a pronounced impact on female reproductive health, including causal contribution to PCOS-related traits like hirsutism and post-menopausal bleeding (PMB). We also illustrate how testosterone levels associate with antagonistic effects on stroke risk and reproductive end-points between the sexes.

**Conclusions** Overall, these findings provide insight into how genetically determined testosterone correlates with several health parameters in both sexes. Yet the lack of evidence for a causal contribution to most traits beyond sex-specific health underscores the complexity of the mechanisms linking testosterone levels to disease risk and sex differences.

#### Plain language summary

Hormones, such as testosterone, travel around the body communicating between the different parts. Testosterone is present at higher levels in men, but also present in women. Variable testosterone levels explain some differences in human traits and disease prevalence. Here, we study how adult testosterone levels relate to health and disease. Genetic, i.e. inherited, differences in testosterone levels contribute to many traits specific to men or women, such as women's reproductive health, hormonal cancers, and hair growth typical in males. However, testosterone levels do not appear as a major cause of most traits studied, including psychiatric diseases and metabolic health. Normal variation in baseline testosterone levels thus seems to have a relatively minor impact on health and disease.

<sup>&</sup>lt;sup>1</sup> Institute for Molecular Medicine Finland (FIMM), HiLIFE, University of Helsinki, Helsinki, Finland. <sup>2</sup> Department of Clinical Chemistry, Finnish Cardiovascular Research Center, Tampere, Finland. <sup>3</sup> Faculty of Medicine and Health Technology, Tampere University, Tampere, Finland. <sup>4</sup> Fimlab Laboratories, Tampere, Finland. <sup>5</sup> Department of Clinical Physiology, Finnish Cardiovascular Research Center, Tampere, Finland. <sup>6</sup> Centre for Population Health Research, University of Turku and Turku University Hospital, Turku, Finland. <sup>7</sup> Research Centre of Applied and Preventive Cardiovascular Medicine, University of Turku, Turku, Finland. <sup>8</sup> Department of Clinical Physiology and Nuclear Medicine, Turku University Hospital, Turku, Finland. <sup>9</sup> Department of Obstetrics and Gynecology, PEDEGO Research Unit, Medical Research Centre, Oulu University Hospital, University of Oulu, Oulu, Finland. <sup>10</sup> Broad Institute of MIT and Harvard, Cambridge, MA, USA. <sup>11</sup> Department of Endocrinology, Abdominal Centre, Helsinki University Hospital, Helsinki, Finland. <sup>12</sup> Folkhalsan Research Center, Helsinki, Finland. <sup>13</sup> Research Programs Unit, Clinical and Molecular Metabolism, University of Helsinki, Helsinki, Finland. <sup>14</sup> Lund University Diabetes Centre, Department of Clinical Sciences Malmö, Lund University, Malmö, Sweden. <sup>15</sup> Department of Public Health, Faculty of Medicine, University of Helsinki, Helsinki, Finland. <sup>16</sup> Helsinki Institute for Information Technology HIIT and Department of Mathematics and Statistics, University of Helsinki, Finland. <sup>26</sup>These authors contributed equally: Nina Mars, Leevi E. Lehtonen, Ari Ahola-Olli. \*A list of authors and their affiliations appears at the end of the paper. <sup>80</sup>email: taru.tukiainen@helsinki.fi

estosterone (T) is the male sex hormone responsible for regulation of development of primary and secondary male sexual characteristics. Individual variation in T levels has been suggested to shape human physiology broadly, including effects on disease risk in both men and women<sup>1–3</sup>. Epidemiological studies and randomized clinical trials for T replacement therapy have observed associations between serum T levels and various traits ranging from type 2 diabetes (T2D) and cardiovascular disease to body composition and behavior<sup>1–11</sup>. Yet, these studies have yielded partly mixed results, and, in many instances, the proposed relationships between T, complex traits and disease remain elusive<sup>2,7–11</sup>.

Besides the disease links, T is a known driver for sex differences. After puberty, men and women differ extensively with respect to their average T levels, with males showing roughly 7–15-fold higher serum total T concentrations<sup>1,12</sup>. This difference largely results from the testicular T production in males that far exceeds the amount of T produced in the ovaries and the adrenal gland in females, and is known to directly contribute to variation in, for instance, body composition between the sexes<sup>1,12</sup>.

In the human body, the majority of T is bound to a carrier molecule, whereas only a small fraction (1-3%) of this total T exists as free T<sup>13-16</sup>. Free T is considered to represent the most potent form of T in terms of biological activity, and although extensively debated may have different clinical significance than total T<sup>13-16</sup>. Most of the remaining T in circulation is tightly bound by sex hormone-binding globulin (SHBG), and the bulk of the rest remains attached to carrier proteins like serum albumin (Fig. 1a)<sup>4,13,14,16</sup>.

Complex physiological processes regulate circulating T levels, allowing T levels to fluctuate based on internal and external stimuli on a daily basis<sup>17,18</sup>. However, twin studies indicate that heritability

of serum T is relatively high in both sexes, being up to 65% in males<sup>19,20</sup>. Until recently, few genetic variants were shown to associate with circulating T levels and related traits in the general population<sup>4,19,21,22</sup>. Yet, through the emergence of genetic and biomarker data from large biobanks, currently more than a hundred, mostly autosomal, loci for serum total T, SHBG and free T have been identified, with evidence for sex-specific genetic effects<sup>23–25</sup>.

Recent efforts have used genetics to address the potential causal contribution of adult T to selected complex traits in both sexes, including T2D, body composition and hormonal cancers, or have examined the effects of free T more broadly in males based on the UK Biobank data<sup>23,24,26–28</sup>. Here, we extend these investigations combining data from the UK Biobank (N = 408,186) and Finn-Gen (N = 217,464). Leveraging the extensive healthcare registry data of FinnGen, sex-specific polygenic scores (PGS), and public data from genome-wide association studies (GWAS), we studied the role of T levels in both sexes across traits ranging from metabolic conditions and sex-specific reproductive disorders to neurological and behavioral endpoints.

We illustrate that normal variation in genetically set T levels has only subtle effects to most human traits. For example, the association between T and many metabolic phenotypes may instead largely stem from shared etiology, challenging some previous findings. Yet we stress the unique genetic architecture of serum T levels, potentially reflecting sexually antagonistic effects of T on fitness, and emphasize the causal role of the hormone also for women's health.

#### Methods

Genotype and phenotype data from the UK Biobank. The genetic association analysis was based on data from the UK



**Fig. 1 Illustration of the studied traits and study overview. a** The traits subjected to GWAS (total T, SHBG and free T) and distribution of these in the UK Biobank. The barplots illustrate division of serum total T to SHBG-bound (brown), and bioavailable fractions (green) including free T (bright green) for men (dark green) and women (pink). The three boxplots in **a** show sex-specific distributions for serum total T, SHBG and calculated free T levels for the UK Biobank participants included in the GWAS. These boxplots show median for each trait (black line), lower and upper quartiles (colored area of the box) and the error bars indicate 5 and 95% quantiles. **b** Overview of the study design to assess the contribution of T to health and disease using genetic approaches and biobank data. We conducted the discovery GWAS in the UK Biobank (N = 408, 186), built sex-specific PGSs for the three T-related traits and validated the predictive ability of the PGSs in the Young Finns Study (YFS) (N = 2055). We then performed complex disease and trait associations using the PGS in FinnGen (release 5, N = 217, 464), complementing these analyses with Mendelian Randomization analyses and extending these analyses with publicly available GWAS data from other cohorts.

Biobank, a population-based biobank consisting of 502,637 subjects (aged 37–73 years)<sup>29</sup>. Approximately 74% of the female participants in the cohort have reportedly undergone menopause. At recruitment, participants provided electronic signed consent. Ethics approval for the UK Biobank study was obtained from the North West Centre for Research Ethics Committee (11/NW/ 0382). All experiments were performed in accordance to relevant guidelines and regulations including the Declaration of Helsinki ethical principles for medical research. This study was run under UK Biobank application number 22627.

Multiple biochemical assays have been performed on the entire UK Biobank cohort, including measurements of serum testosterone, SHBG and serum albumin. These measurements were performed once for each participant for using Beckman Coulter DXI 800 Chemiluminescent Immunoassay, with competitive binding for testosterone and two-step sandwich for SHBG, referenced in https://biobank.ndph.ox.ac.uk/showcase/showcase/ docs/serum biochemistry.pdf. The document contains also details about the extensive QC procedures for all biochemical measurements in the UK Biobank. Analytical range for the immunoassays of total testosterone and SHBG were 0.35 to 55.52 and 0.33 to (226-242) nmol/L, respectively. Assay consistency was monitored by using internal QC samples between batches, and external quality assurance schemes against the ISO 17025:2005 standard. Notably, the reported testosterone and free T levels in the UK biobank are on average lower than for many other cohorts, for example YFS (Supplementary Data 2 and 6)<sup>23,26</sup>, which may be partly accountable for UK Biobank measures being based on immunoassays instead of mass spectrometry.

We restricted our study to encompass 408,186 individuals from the white British subset. We removed outliers for genotype heterozygosity and missingness, as well as samples with sex chromosome aneuploidies, mismatches between reported and inferred sex, and samples that UK Biobank did not use in relatedness calculations<sup>29</sup>. We included related samples since our analysis method (BOLT-LMM) allows for their inclusion. To ensure that the UK Biobank measurements capture baseline heritable variation in adult T levels, we assessed correlation of testosterone measurements between two timepoints available for a subset of a UK Biobank population (7097 men and 5285 women included in the primary analyses), with the original samples collected between 2006 and 2010, and additional samples taken in 2012–2013. T levels from these two time points showed a high correlation (R = 0.678 for males and R = 0.709 for females, Supplementary Fig. 3).

For the GWASs, we used biochemically measured testosterone and SHBG, and calculated FAI and free T. For clarity, given the close relationship between free T and FAI, in the results section we concentrate on describing the results for free T, whereas the results for FAI are available in the supplementary data. For calculation of FAI, we used the formula *100\*Testosterone/SHBG* (nmol/ml). Calculated free T was derived using the Vermeulen equation using directly measured albumin values for each participant<sup>30,31</sup>.

All four traits were separated by sex and log transformed. Subjects with residuals values of +/-5 SD from the mean were excluded from the analyses, serving as a further QC step to exclude outliers potentially reflecting medical conditions or drug use affecting androgen levels. Inverse normalized values of the remaining residuals were used as phenotype values for the GWAS analyses. After the QC steps, our study included altogether 177,499 men and 205,141 women.

Genetic association analysis and definition of the lead SNPs. The GWAS analyses were performed using BOLT-LMM  $(v2.3.2)^{32,33}$ . Imputed SNPs were restricted to variants with

MAF  $\ge 0.1$  % and imputation quality  $\ge 0.7^{29}$ . We included both autosomal and X-chromosomal variants into the analysis, excluding pseudoautosomal regions (PARs). X-chromosomal effects in males were based on (0,2) allele dosage coding. 1000 Genomes European data was used as reference LD scores for calibrating the BOLT-LMM statistic. First 10 Principal Components were used as quantitative covariates in the runs. A linear regression model was fit for each trait with BMI and age as covariates, as well as menopause status for females. Genetic correlation analyses, heritability estimates and number of loci found implied the results remained consistent with different covariate configurations (Supplementary Fig. 10)<sup>28,29</sup>. Including up to 127 covariates (based on ref. <sup>34</sup>), e.g., assay center, dilution factors, blood draw time, and socioeconomic status indicators, or excluding related individuals from the analysis all showed negligible effects on the genetic findings (Supplementary Fig. 10).

Independent lead SNPs were selected for each chromosome by recursively taking the SNP with the lowest p-value (until none below the p-value threshold 5e-08 were left) from the GWAS summary statistics and removing all SNPs 500 kb on each side of it from the next round. The chromosomal positions of these 1 mb windows were stored, and overlapping windows were merged into the final list of loci. The SNP with the lowest p value in each of these windows was selected as the lead SNP.

Statistics and reproducibility. Details of the variety of statistical methods used are described in the methods subsections found below. Our study design comprised of several steps to ensure robustness of the findings, and the conclusions of the study are based on multiple converging statistical analyses and vast datasets. In short, our study included confirming the reproducibility of the genetic findings from the UK Biobank (N = 408,186) (Supplementary Data 1) using the Young Finns Study (N = 2,055) (Supplementary Data 6), running discovery PGS and MR analyses in FinnGen (N = 217,464) (Supplementary Data 7 and 8) and replicating and extending these findings using publicly available GWAS summary statistics with data from tens of thousands to over a million participants (Supplementary Data 11–13).

**Pathway, tissue enrichment and co-localization analyses**. Tissue and gene set enrichment analyses were carried out with SNP2GENE and GENE2FUNC implemented in FUMA using default settings<sup>35</sup>. For testing in which tissues the genes residing in the GWAS loci were preferably expressed, we used the full distribution of SNP p-values and the GTEx v6 30 general tissue types. For pathway analysis, to assess whether the genes in the GWAS loci are overrepresented in pre-defined gene sets via hypergeometric tests, we selected manually curated KEGG-pathways<sup>36</sup>. For co-localization analyses to assess whether the genetic loci showed evidence for shared genetic effects between males and females, and to estimate the maximum posterior probability (MAP) for the loci being shared, we used gwas-pw<sup>37</sup>.

**Replication in the Young Finns Cohort and calculation of PGS.** The Cardiovascular Risk in Young Finns Study (YFS) is a longitudinal follow-up of 3,596 subjects at baseline. The baseline survey was conducted in 1980 and subsequent follow-ups involving the whole sample were held in 1983, 1986, 2001, 2007, 2011, and 2017. The YFS study has been approved by the ethical committee of the Hospital District of Southwest Finland, all study subjects gave an informed consent, and the study has been conducted according to the principles of the Declaration of Helsinki. For testosterone and SHBG, we used data on 2001 follow-up (Subjects aged 24–39 years). A venous blood sample was drawn from antecubital vein after 12-h overnight fast. Serum was aliquoted and stored at -70 degrees Celsius until analysis. In males, total testosterone quantification was performed in 2009 with competitive radioimmunoassay (Spectria Testosterone kit, Orion Diagnostica, Espoo, Finland) and Bio-Rad Lyphocheck control serums 1, 2, and 3 were used in quality control. Before quantification, serum aliquots had been melted three times. Total testosterone was quantified first and aliquots were re-frozen before SHBG quantification. In females, total testosterone quantification was performed in 2011. SHBG quantification for males was done in 2009 and for females in 2011 with Spectria SHBG IRMA kit (Orion Diagnostica, Espoo, Finland). Free testosterone was estimated using Vermeulen's formula. As albumin concentration was not available, we used fixed albumin concentration of 43 g/l.

In YFS, genotyping was performed at the Wellcome Trust Sanger Institute (UK) using customized Illumina Human Map 670k bead array. The custom content on 670k array replaced some poor performing probes on Human610 and added more CNV content. As quality control, we excluded individuals and probes with over 5% of missingness (--geno and --mind filters in Plink). Variants deviating from Hardy-Weinberg equilibrium ( $p < 1 \times 10^{-6}$ ) and minor allele frequency below 1% were excluded. Related samples were excluded (n = 51) with pi-hat cut-off of 0.2. Total of 2442 individuals and 546,674 variants passed the quality control measures. The mean call rate across all included markers after the quality control was 0.9984. Next, imputation was performed using population-specific Sequencing Initiative Suomi (SISu) as reference panel. We examined the association of the UK Biobank GWAS lead SNPs with the corresponding T trait in YFS. If the annotated lead SNP was not available in YFS, we used LDstore2 (v2.0b)<sup>38</sup> to calculate LD in a 100 kb window around the lead SNP in the UK Biobank imputed data and selected the closest SNP with  $R^2 > 0.8$  with the lead SNP as a proxy.

To construct PGSs we applied the LDpred<sup>39</sup> method to the sexspecific GWAS summary statistics from the UK Biobank for total T, SHBG, FAI and free T, using 1000 Genomes Europeans as LD reference and the default LD radius to account for LD. We then used the weights from the LDpred infinitesimal model to construct genome-wide PGSs for each individual in the YFS with Plink 2.0 (11 Feb 2018). Only variants imputed with high confidence (imputation INFO > 0.8) were included in PGS calculation. Variants in chromosomes 1–22 and chrX were included. In males, allele dosage of 2 was used for X-chromosomal haploid variants. To evaluate the PGS prediction accuracy in the YFS, we calculated the  $R^2$  for each trait using using linear regression with z-score normalized PGS as predictor, age and 10 PCs as covariates and z-score normalized T trait as outcome.

Estimation of heritability and calculation of the genetic correlations by LDSC. SNP-based heritability for the studied T traits and genetic correlations between these and with 44 additional phenotypes were estimated using linkage disequilibrium score regression (LDSC), including autosomal variants only<sup>40</sup>. The summary statistics for the 44 traits were downloaded directly from the source repositories and analyzed locally, for the original sources please see references in Supplementary Data 11. For the genetic correlation analyses in LDSC, pre-computed LD Scores from 1000 Genomes Europeans, excluding the HLA region were used. For 23 traits we performed the analyses using sex-specific GWAS results and compared these to data from sex-combined GWAS (Supplementary Data 13). Generally, genetic correlation results using either sex-specific or sex-combined GWAS data were highly similar.

**Disease associations in FinnGen**. To assess if the PGS for studied traits associate with disease risk we utilized the FinnGen study

(data freeze 5), consisting of 217,464 (94,478 men, 122,986 women)<sup>41</sup>. FinnGen is comprised of Finnish prospective epidemiological and disease-based cohorts and voluntary biobank samples collected by hospital biobanks. The genotypes have been linked to national hospital discharge (available from 1968), death (1969-), cancer (1953-) and medication reimbursement (1964-) registries as well as the registry on medication purchases (1995-). Although specific menopause status for the participants is not available, roughly 67% of the females in R5 (N = 84,521) are estimated to have undergone menopause at the end of the follow-up (age >50). The samples were genotyped with Illumina and Affymetrix arrays (Illumina Inc., San Diego, and Thermo Fisher Scientific, Santa Clara, CA, USA). The genotypes have been imputed with using the SISu v3 population-specific reference panel developed from high-quality data for 3775 high-coverage (25-30×) whole-genome sequencing in Finns. The detailed genotype imputation workflow can be found at https://doi.org/10. 17504/protocols.io.xbgfijw. The dataset uses genome build 38 (hg38). A full list of FinnGen consortium members can be found from Supplementary Information.

For PGS analyses, we used same variant weights (LDpred infinitesimal model) as for YFS, and calculated genome-wide PGSs for each individual with PLINK2 (v2.00a2.3LM). Variants in chromosomes 1-22 and chromosome X (imputed with high confidence, imputation INFO  $\geq$  0.7) were included (total number of variants ranging from 6,535,263 for female total T to 6,536,405 for female SHBG) and we used genotype dosages to incorporate imputation uncertainty. In males, allele dosage of 2 was used for X-chromosomal haploid variants. We studied the PGS associations to 36 disease endpoints with potential links to androgens, representing six loosely defined disease categories. For details of the studied phenotypes see Supplementary Data 7, www.finngen.fi and risteys.finngen.fi. Cox proportional hazards models were used for estimating hazard ratios (HRs) and 95% CIs, with age as the time scale and 10 first principal components of ancestry and genotyping batch as covariates. The proportionality assumption for Cox models was assessed with Schoenfeld residuals and log-log plots. We performed power analyses on the Cox models using the power-EpiCont.default from powerSurvEpi package in R. We estimated that with the selected *p*-threshold our study is well-powered to reliably detect large effects (Hazard Ratio (HR) > 1.3) the PGSs have on rare endpoints such as hirsutism, and smaller effects (HR > 1.1) on the vast majority of the studied diseases (>1800 cases, 29 traits in both sexes, Supplementary Data 7 and Supplementary Fig. 7). For the cross-sex analyses, we took the sex-specific PGSs, and checked whether these would associate with the studied endpoints in the other sex, using the z-test to compare equality between the original and cross-sex associations. We additionally performed SHBGadjusted PGS associations to all endpoints to control for the effects of SHBG on total and free T associations. Although in both the PGS and causality analyses we opted for adjusting for the effects of SHBG, BMI, PCOS and menopause in women-(Supplementary Data 7-12 and Supplementary Fig. 10)<sup>42-45</sup>—we remind that related factors may still confound our findings.

Patients and control participants in FinnGen provided informed consent for biobank research, based on the Finnish Biobank Act. Alternatively, older research cohorts, collected prior the start of FinnGen (in August 2017), were collected based on study-specific consents and later transferred to the Finnish biobanks after approval by Valvira, the National Supervisory Authority for Welfare and Health. Recruitment protocols followed the biobank protocols approved by Valvira. The Coordinating Ethics Committee of the Hospital District of Helsinki and Uusimaa (HUS) approved the FinnGen study protocol Nr HUS/990/2017.

The FinnGen study is approved by Finnish Institute for Health and Welfare (THL), approval number THL/2031/6.02.00/2017,

amendments THL/1101/5.05.00/2017, THL/341/6.02.00/2018, THL/2222/6.02.00/2018, THL/283/6.02.00/2019, THL/1721/5.05.00/2019, Digital and population data service agency VRK43431/2017-3, VRK/6909/2018-3, VRK/4415/2019-3 the Social Insurance Institution (KELA) KELA 58/522/2017, KELA 131/522/2018, KELA 70/522/2019, KELA 98/522/2019, and Statistics Finland TK-53-1041-17. The Biobank Access Decisions for FinnGen samples and data utilized in FinnGen Data Freeze 6 include: THL Biobank BB2017\_55, BB2017\_111, BB2018\_19, BB\_2018\_34, BB\_2018\_67, BB2018\_71, BB2019\_7, BB2019\_8, BB2019\_26, Finnish Red Cross Blood Service Biobank 7.12.2017, Helsinki Biobank HUS/359/2017, Auria Biobank AB17-5154, Biobank Borealis of Northern Finland\_2017\_1013, Biobank of Eastern Finland 1186/2018, Finnish Clinical Biobank Tampere MH0004, Central Finland Biobank 1-2017, and Terveystalo Biobank STB 2018001.

Causality analyses and their limitations. MR analyses treat genetic variants as instrumental variables and their reliability depends on two key assumptions: (1) alleles are randomly assigned, and (2) that alleles that influence exposure do not influence the outcome via any other means. The first assumption is partly controlled by using BOLT-LMM as our model in the primary GWAS analysis together with using PCs to control for population stratification. Although demographic phenomena such as assortative mating (mate choice based on similar characteristics in spouse) may affect effect estimates for measures such as educational attainment, based on within-sibship GWAS<sup>46</sup> biochemical traits such as T levels seem to less affected by such confounders. The second limitation (especially in the form of genetic pleiotropy) is harder to control when using a large number of SNPs as instrumental variables. Susceptibility to confounding by pleiotropy (a gene influences multiple traits via independent biological pathways<sup>47</sup>), overall presents a special issue for genetic studies.

In this case, in line with the observed wide-spread genetic pleiotropy affecting most complex traits, we noted that the GWAS loci contained many genes associated with pleiotropic effects on human phenotypes (for example,  $LIN28B^{48}$ ,  $GCKR^{37}$  and  $TYK2^{49}$ ). The cross-sex analyses further supported the notion that such pleiotropic variants (for example, missense variants in genes like *GCKR*, *SERPINA1*, and *SLCO1B1*, associating with T and SHBG levels but having other biological functions (Supplementary Data 1)<sup>50–52</sup>) may partly underlie, e.g., T's correlations with T2D and hypothyroidism.

Therefore, given the vast polygenicity of the studied T traits, we chose latent causal variable (LCV)<sup>53</sup> and MR-Egger<sup>54</sup> as our primary causality analysis methods, designed to take into account pleiotropy-induced confounding. LCV has been proposed to provide more unbiased causality estimates than conventional MR approaches, whereas MR Egger should provide accurate causality estimates under the InSIDE assumption (the genetic variants have pleiotropic effects that are independent in magnitude and are thus not mediated by a single confounder exposure), besides its recommended use as a sensitivity analysis for conventional MR<sup>53,54</sup>. For T traits, we additionally compared the causality results of these models to estimates from CAUSE, a MR method similarly designed to distinguish pleiotropy from true causality (Supplementary Data 2)<sup>55</sup>. In the analysis of other complex traits, for comparison we also ran conventional MR analyses (Inverse-Variance Weighed (IVW)), that remains sensitive for confounding by genetic correlation and pleiotropy<sup>53</sup>. To extend the basic MR Egger analysis and to tease out the potential effects of SHBG on causality estimates of total and free T, we used multivariable MR Egger<sup>56</sup>. LCV reports genetic causality proportion (GCP) as an estimate of causality, under a model where genetic correlation between two traits is mediated by a latent variable having a causal effect on each trait. GCP = 1 means trait 1 is fully correlated with the latent variable, and hence fully causal to trait 2. A high GCP value and a statistically significant effect support partial genetic causality between the traits, and suggest that interventions targeting trait 1 are likely to affect trait 2. The *p* value obtained in the analysis refers to the null hypothesis that the GCP = 0. A highly significant p value does not require a high GCP, but in the presence of lower GCP suggests only partial causality between the traits. Positive GCP value indicates causality of trait 1 to trait 2, whereas a negative value indicates support for causality of trait 2 to trait 1. LCV also estimates genetic correlation between the traits. Notably, LCV assumes a single latent variable mediating the effect between two phenotypes, and that the effect can be only unidirectional. Hence it may be confounded by bidirectional causal effects or by the presence of several latent variables. Given these limitations of the methods, despite the vast potential of MR in establishing causal relationships<sup>57</sup>, we however propose caution in interpreting these results: also in our case the MR models did not always agree on causality, common to recent MR studies assessing the function of  $T^{23,24,26}$ .

To estimate the how much T could explain sex differences in hemoglobin we calculated ((FT\_m-FT\_f/SD\_FT\_m)\*\beta\*SD\_H\_m)/ ((H m-H f)/SD H m), where FT = mean free Testosterone, m = males, f = females, SD = standard deviation,  $\beta = MR$  estimate, H = Hemoglobin in UK Biobank based on ref. <sup>58</sup>. We applied the LCV, MR-Egger, CAUSE, multivariate MR-Egger and IWV models locally using R 4.0.2. The MR analyses were run using TwoSampleMR (v0.5.2)<sup>59</sup> and MendelianRandomization (v0.5.0) R packages<sup>60</sup>. These causality analyses on the FinnGen traits included both autosomal and X-chromosomal SNPs. For all causality analyses X-chromosomal effects in males are presented for (0,2) allele dosage coding. For the traits from public GWAS included in genetic correlation analyses, in 16 out 44 instances we could perform two sample MR (phenotype data not including UK Biobank samples) and in 19/44 instances the studies included X-chromosomal data (Supplementary Data 11). FinnGen represents an independent research cohort from the UK Biobank and thus all FinnGen causality analyses were two-sample MR analyses.

**Reporting summary**. Further information on research design is available in the Nature Portfolio Reporting Summary linked to this article.

#### Results

We utilized the rich biochemical and health information available in two population-scale genetic datasets and analysis methods building on GWAS discovery. In brief, we conducted sex-stratified GWAS for T, SHBG and free T based on immunoassay measurements from the UK Biobank, and constructed sex-specific polygenic scores (PGS) for these traits. The PGSs capture the genetic effects on T and SHBG levels, and therefore serve as a proxy for cumulative post-pubertal T exposure. Using an external dataset (Young Finns Study; YFS), we validated the performance of the PGSs. We then investigated the effects of the PGSs on a wide range of diseases across diverse clinical entities using the FinnGen study. Lastly, we evaluated causal relationships and genetic correlations between the studied T traits and complex traits, leveraging publicly available GWAS summary statistics (Fig. 1).

**GWAS and polygenic scores for testosterone traits**. We identified more than a hundred genome-wide significant (p < 5e-08) loci the autosomes and the X chromosome combined for all testosterone traits (up to 263 loci for SHBG in males) in the UK Biobank GWAS (Methods; Supplementary Data 1). In both sexes, common variants (allele frequency >1%) covered a large proportion of the trait

variability, with the SNP heritability ( $h^2$ ) estimates ranging from 10% for total T in females to 28% for SHBG in males (Supplementary Data 2). The associated loci were enriched for genes affecting steroid hormone biosynthesis, metabolism and excretion, with preferential expression in the liver for all the studied traits, in line with recent findings (Supplementary Figs. 1 and 2 and Supplementary Data 3)<sup>23–25</sup>. Reflecting the high heritability, T measurements from two different time points were highly correlated in both sexes (Supplementary Fig. 3). Collectively, despite being based on immunoassays which may have limited use in clinical settings especially in females<sup>61,62</sup>, these findings illustrate that the UK Biobank data permits construction of robust genetic instruments to study how post-pubertal T and SHBG levels relate to adult health.

The loci affecting SHBG were largely shared between the sexes (genetic correlation  $(r_g) = 0.88$ , p = 9.7e-197), whereas we observed a near-zero genetic correlation estimates for both serum T and free T between males and females ( $r_g = 0.08$  and 0.05, respectively, p > 0.05), indicating sex-specific genetic determinants, as previously reported<sup>23,24</sup>. Co-localization analyses between male and female GWASs confirmed the widespread sex-specificity of the genetic loci for T (Supplementary Fig. 4, Supplementary Data 4 and Supplementary Data 5)<sup>24</sup>.

Reflecting the sex-specific genetic architecture, we observed strong genetic correlation between total T and SHBG only in males ( $r_g = 0.78$  in males vs. 0.05 in females, Supplementary Fig. 4 and Supplementary Data 2). In females, instead, the genetic determinants for free T were shared with SHBG ( $r_g = -0.65$ ). Correspondingly, SHBG was found causal for total T levels in males (genetic causality proportion (GCP) = 0.80, p = 5.8e-05), whereas in females SHBG appeared to control the free T fraction (GCP = 0.83, p = 3.8e-07).

To study the impacts of T and SHBG levels in datasets where these measurements are not directly available, we next constructed sex-specific genetic predictors, PGSs, for each trait applying the LDpred algorithm<sup>39</sup> to the sex-specific GWASs. We tested the predictive ability of the PGSs in the YFS where the phenotypic variance explained by a genome-wide PGS ( $R^2$ ) ranged between 1.1% (male free T) to 9.2% (male SHBG), indicating the PGSs predict T and SHBG levels in an independent cohort (Supplementary Data 6). Notably, the sex-specific PGSs for T and free T had no predictive value in the opposite sex, as expected based on serum T being determined by distinct genetic variants in both sexes (Supplementary Data 6 and Supplementary Fig. 5).

Studying the links between cumulative T exposure and disease. We continued by using the PGS to study how post-pubertal T exposure associates with disease risk, the associations potentially implying causal relationships<sup>28</sup>. In short, these analyses allow for estimating the consequences of having a genetic predisposition to higher or lower T levels (Supplementary Fig. 6). To this end we used the FinnGen data, consisting of 217,464 (94,478 men, 122,986 women) Finnish participants, representing roughly 5% of the Finnish adult population, with genotypes linked to up to 46 years of follow-up within nationwide healthcare registries<sup>41</sup>. We studied 36 diseases with potential links to hormones from the following categories: (1) endocrine and metabolic diagnoses, (2) sex-specific endpoints (many specific to females, e.g., postmenopausal bleeding, (PMB)), (3) cardiovascular and circulatory system, (4) nervous system disease, (5) behavioral and neurological diagnoses and (6) other endpoints like injury risk (Supplementary Data 7). The number of cases ranged from 229 individuals diagnosed with hirsutism to 68,774 statin users. Power analyses supported our ability to detect large effects (HR > 1.3) even for the rarest, and smaller effects (HR > 1.05) on more common phenotypes (Supplementary Fig. 7 and Methods).

We first tested whether the PGSs are associated with disease risk in a sex-specific manner, observing 32 associations (p < 0.0014, after Bonferroni correction for 36 independent tests; Supplementary Data 7 and Supplementary Fig. 6). The associations involved endocrine, metabolic and sex-specific disorders, highlighting in particular female-specific endpoints (Fig. 2 and and Supplementary Data 7). In men, both higher total T and SHBG PGSs usually associated with reduced disease risk. In women, both higher total T and free T PGSs generally increased risk for multiple diseases, often showing inverse associations to SHBG.

Underscoring T's and SHBG's involvement in metabolism, larger PGS values for total T and SHBG were associated with reduced T2D risk and statin use in men (for total T in males, HR = 0.94, p = 1.4e-17 and 0.96, p = 1.1e-15, respectively). In contrast, higher free T PGS in women increased risk for both (HR = 1.09, p = 1.9e-22 and HR = 1.02, p = 0.0011). Given the shared association profiles with SHBG, we included SHBG as a covariate in complementary analyses to distinguish true T-driven effects. These suggested that associations to metabolic traits were not primarily attributable to androgen action (Supplementary Data 7 and Supplementary Fig. 8). In women, higher T PGS was additionally associated with lower hypothyroidism risk (HR = 0.97, p = 6.5e-05), persisting SHBG adjustment (HR = 0.97, p = 8.6e-05). We also detected suggestive associations to bone strength and injury risk in both sexes, but with the exception of SHBG and osteoporosis in females (HR = 1.08, p = 0.00015), none of these findings survived correction for multiple testing.

In the sex-specific category, we replicated the known associations of T to PCOS and breast cancer risk in women<sup>23</sup> (HR = 1.02, p = 2.8e-06 and HR = 1.04, p = 0.0001 for free TPGS) (Fig. 2 and Supplementary Data 7). Of the previously unstudied endpoints, we robustly linked T with hirsutism and post-menopausal bleeding (PMB) (HR = 1.45, p = 2.7e-08 and HR = 1.05, p = 0.00032 for free T PGS), with connections to PCOS, and endometrial cancer, respectively<sup>63,64</sup>. The association to hirsutism (excessive hair growth in a male-type fashion) appeared particularly pronounced, with the risk of the condition almost doubling with a 2 SD change in T. In FinnGen, there was only modest case and genetic overlap between PCOS, hirsutism and PMB cases, potentially reflecting underdiagnosis of PCOS in the dataset<sup>65</sup>. SHBG adjusted analyses further suggested these associations were androgen dependent (Supplementary Data 7 and Supplementary Fig. 8). In contrast, reflecting the close relationship between SHBG and calculated free T in females, some associations for female free T like infertility risk (HR = 1.04, p = 0.00502), seemed highly dependent on SHBG, showing generally favorable effects on reproductive traits, e.g., HR = 0.98, p = 0.00116 for irregular menstruation).

We observed no statistically significant associations to other diseases (Supplementary Fig. 6), including the 13 neurological/ behavioral endpoints studied, like Alzheimer's disease, alcohol use, and anxiety disorders (all p > 0.0014). However, the PGSs did show some nominal clinically-relevant associations: higher PGSs for male free T was linked with prostate cancer and lower osteoporosis incidence (HR = 1.03, p = 0.0083 and HR = 0.89, p = 0.0023, respectively). Male total T associated with stroke (HR = 0.97, p = 0.017), and female total T with cardiac death (HR = 1.04, p = 0.049) and Parkinson's (HR = 1.08, p = 0.017). Both male free T and female total T PGS showed associations to lower anemia risk (HR = 0.97, p = 0.042 HR = 0.98, p = 0.041, respectively, Fig. 2 and Supplementary Data 7).

**Evaluation of causal relationships between T and disease**. Although the PGS associations may imply direct causality of the first trait to another, these are also prone for confounders



**Fig. 2 Results from the PGS associations with disease endpoints in the FinnGen. a** Illustrates the distribution of association *p*-values by disease category for each biomarker PGS separately for both sexes. For clarity *p* values capped at 1e-20. Dark green symbol = males, pink = females. Green circles = total testosterone, white lilac = SHBG, blue = free testosterone. **b** Shows hazard ratios per one SD increase in PGS for 20 traits from endocrine, metabolic, circulatory and sex-specific categories. Yellow shades = increased risk for endpoint, blue shades = decreased risk for endpoint. The data is based on 94,478 men and 122,986 women from FinnGen R5. Case and control numbers for each endpoint are available from Supplementary Data 7. \**p* < 0.05, \*\**p* < 0.0014, corresponding to Bonferroni correction for 36 independent traits. PCOS polycystic ovary syndrome, PMB post-menopausal bleeding, CHD coronary heart disease.

including genetic pleiotropy<sup>47</sup>. We estimated causal relationships between T and the studied endpoints using two complementary MR methods that aim to correct for potential pleiotropy: LCV<sup>53</sup>, representing a genome-wide approach, and MR-Egger<sup>54</sup> that uses significantly associated SNPs (Methods). For the sex-specific PGS data, in 23/252 instances one or both methods suggested evidence for a causal relationship (p < 0.0014) between a PGS and a disease (Fig. 3, Supplementary Fig. 9, and Supplementary Data 8). Finally, we distinguished between the effects of T and SHBG by including the latter as a covariate in multivariable MR Egger analyses<sup>56</sup>.

Despite the many associations (Fig. 2), also in the MR analyses we observed no clear evidence for T being directly causal for metabolic traits. We consistently saw no significant causality between T and T2D, contrasting some recent findings (Fig. 3, Supplementary Fig. 9,



**Fig. 3 Results from causality analyses in FinnGen, showing MR estimates for total and free T in males and females.** The figure includes traits from the endocrine, metabolic, circulatory and sex-specific categories shown in Fig. 2. The data is based on 94,478 men and 122,986 women from FinnGen R5. Case and control numbers for each endpoint are available from Supplementary Data 8. Dark green symbol = males, pink = females. Estimates shown in green = total testosterone, blue = free testosterone. Shown are both genetic causality proportion (GCP) estimates from LCV (within the gray boxes, 0 = no causality, 1 = fully causal, minus sign indicates suggested reverse causality, i.e., the endpoint affecting T), and MR Egger betas and standard error for each trait (horizontal lines). The MR Egger value in the x-axis corresponds to 1SD increase in risk per 1SD increase in T/free T. \*p < 0.05, \*\*p < 0.0014, n.a. = not applicable due to low heritability estimate in FinnGen under the LCV model.

and Supplementary Data 8)<sup>23</sup>. In addition, the partly surprising links between total T, higher risk for osteoporosis and lower risk for statins seen in standard MR in males appeared attributable to SHBG.

In contrast, the causality analyses strongly supported the role of T in the regulation of female reproductive health. For instance, MR Egger supported causality of total T ( $\beta = 0.90$ , p = 0.012) and LCV of free T to PCOS (GCP = 0.54, p = 0.0017) (Fig. 3 and Supplementary Data 8), independently of SHBG. We observed causality also between T, PMB and hirsutism ( $\beta = 0.61$ , p = 4.5e-05and  $\beta = 2.11$ , p = 0.002, respectively, for free T in multivariable MR Egger). Here, adjusting for PCOS had negligible effects on MR results, suggesting these phenotypes are primarily dependent on direct androgen load, and not on metabolic disease (Supplementary Data 8). Additionally, both approaches showed causality between T and hormonal cancers (e.g. GCP = -0.55, p = 0.041 and  $\beta = 0.34$ , p = 0.0031 for female free T and breast cancer; GCP = -0.37, p = 2.8e-16;  $\beta = 0.32$ , p = 0.014 for male free T and prostate cancer). Notably, genetically predicted free T was linked with cancer risk also in the opposite sex (GCP = -0.37, p = 6.6e-19 for male free T and breast cancer, and GCP = -0.74, p = 0.0041 for female free T and prostate cancer).

Lastly, some causal relationships were detected despite no robust associations in the PGS analyses, like the protective association of female T to seropositive rheumatoid arthritis (RA) (Multivariable MR Egger  $\beta = -0.43$ , p = 0.0079). T and SHBG were linked with injury risk in both sexes (e.g., GCP = 0.53, p = 0.0018 and  $\beta = 0.17$ , p = 1.3e-05 for SHBG increasing forearm/elbow injuries in men). Finally, LCV suggested a causal relationship between higher SHBG and ADHD in both sexes, and higher free T and increased risk for conduct disorder but decreased risk for emotionally unstable personality in men (p < 0.0014, Supplementary Data 8), pointing to potential hormonal involvement in the regulation of neuronal processes.

**Results from the cross-sex PGS associations in FinnGen**. Owing to the unique genetic architecture, the sex-specific PGS for T and free T do not predict the corresponding hormone levels in the opposite sex (Supplementary Data 6). Given this, we reasoned that cross-sex analyses, i.e., analysis of the effect of a sex-specific PGS in the opposite sex, would provide a unique opportunity to assess if the original associations stem from T action, and to detect potential antagonistic effects for the PGSs between the sexes.

Aligning with the results from the MR analyses, 15/20 of nominally significant (p < 0.05) sex-specific associations remained similar (Z-test p > 0.05) in the cross-sex analyses, pointing to genetic pleiotropy rather than T action as the basis for many metabolic associations. For example, both female and male total T

PGSs associated with hypothyroidism risk with a similar effect size in the opposite sex, and all PGS associations to T2D were replicated in the other sex (Fig. 4 and Supplementary Data 9 and 10).

The associations differed (*Z*-test p < 0.05) in the opposite sex for male total T PGS and stroke, female total T PGS and anemia, and male free T PGS, head injury risk and osteoporosis. (Fig. 4 and Supplementary Data 9 and 10). In addition, the effect of male total T PGS on statin use was attenuated in females. Intriguingly, while higher male total T PGSs rather reduced stroke risk in men (HR = 0.97, p = 0.017), this associated with increased risk for stroke in women (HR = 1.03, p = 0.017). This was the only endpoint for which we detected evidence for sexual antagonism. The antagonistic effect strengthened with SHBG adjustment (HR = 0.96, p = 0.011 in men and HR = 1.07, p = 0.0010 in women). The results indicate the genetic effects on stroke risk may be partly sex-specific<sup>66</sup>, with potential interplay from sex hormones.

Finally, the cross-sex analyses implied increased androgen load as a direct contributor to poorer reproductive health in women. For the reproductive endpoints that were associated with female free T PGS, i.e., infertility, PMB, PCOS and hirsutism, the male free T PGS had no predictive power (all p < 0.05, Fig. 4 and Supplementary Data 9 and 10). Nonetheless, the effect sizes were not attenuated in a statistically significant manner for breast and prostate cancers, (p > 0.05, Fig. 4), echoing the results from causality analyses and pointing to a degree of shared genetic risk, irrespective of free T levels, between male and female hormonal cancers.

**Extending the FinnGen discoveries**. We next sought to validate and refine the FinnGen discoveries in additional datasets, extending our analysis to include quantitative traits not available in large numbers in FinnGen. To this end, we used genetic correlation analysis, allowing for estimation of the extent to which two traits share genetic factors<sup>40,67</sup>, followed by causality estimations.

We selected 44 traits with publicly available GWAS summary statistics, identical to (e.g., T2D, breast and prostate cancers) or closely reflecting the studied disease phenotypes (heel bone mineral density (HBMD), mood swings) from FinnGen, adding anthropometric traits to the analyses (Supplementary Data 11). For most female-specific phenotypes studied in FinnGen, including hirsutism and PMB, we had no comparable phenotypes, as there were no published GWAS available.

We found evidence of a significant genetic correlation in 72/352 instances (p < 0.0011, corresponding to Bonferroni correction for 44 tests) (Fig. 5 and Supplementary Data 11). Reflecting the FinnGen PGS associations, the genetic correlations involved traits related to metabolism, including biomarkers, but we detected only few correlations to behavioral traits, and no significant correlations to neurological or temperamental traits (Fig. 5 and Supplementary Data 11). Notably, the behavioral traits showing significant correlations were clearly linked to metabolism (smoking, sleep duration and exercise).

The genetic factors increasing serum total T and SHBG overall promoted a favorable metabolic profile in males. Despite correlating with increased BMI, total T and SHBG were positively correlated to adiponectin, high-density lipoprotein (HDL) and lower waist-to-hip ratio (WHR) ( $r_g > 0.20$ , p < 0.0011) while lowering triglycerides and T2D risk in men ( $r_g < -0.25$ , p < 0.0011). In contrast, higher free T fraction in women associated with negative metabolic effects, including higher WHR ( $r_g = 0.25$ , p = 1.6e-22) and lower HDL cholesterol ( $r_g = -0.18$ , p = 4.2e-05). Relatedly, strong correlations were observed between SHBG and metabolic traits in women, including protective associations to markers of liver damage (alanine transaminase (ALAT),  $r_g = -0.21$ , p = 1.7e-08) and gamma-glutamyl transferase (GGT),  $r_g = -0.16$ , p = 0.0001).

We observed significant genetic correlations to hormonal cancers in both sexes, as expected<sup>23,26</sup>. In men, the genetic factors increasing free T promoted prostate cancer ( $r_g = 0.12$ , p = 0.0004), whereas in women these increased especially the risk of estrogen receptor (ER) positive breast cancer ( $r_g = 0.15$ , p = 9.1e-09). Finally, consistent with the existing genetic, epidemiological and experimental data<sup>1,26</sup>, we found evidence for shared genetic background for T, hemoglobin levels and body fat in both sexes (e.g.,  $r_g = 0.15$ , p = 1.0e-07 and  $r_g = -0.14$ , p = 0.0002 for hemoglobin and body fat, respectively, with male free T).

In the LCV and MR-Egger analyses, we found statistically significant evidence (p < 0.0011) of a causal relationship in 7% (26/ 354) of instances across the 44 traits (Fig. 5 and Supplementary Data 12). Reflecting the results from FinnGen, for most traits there was no evidence for substantial causality, but the few suggested causal relationships involved traits with clear biological links to T function. As examples of expected causal relationships<sup>1,26,68</sup>, our analyses supported the contribution of free T levels to male-pattern baldness (MPB, GCP = 0.44, p = 7.7e-19) and hemoglobin levels (GCP = 0.64, p = 0.00085) (Fig. 5 and Supplementary Data 12). In addition, higher T in men was linked to increased number of children fathered (GCP = 0.62, p = 1.2e-15 for total T). The MR Egger analyses supported the causality of female T to ER+ breast cancer ( $\beta = 0.282$ , p = 0.00014 for total T), whereas prostate cancer risk was linked to SHBG levels (GCP = -0.64, p = 0.00019). Additionally, SHBG decreased lymphocyte count in both sexes (GCP = 0.39, p = 1.6e-07 for men, GCP = 0.42, p = 0.00014 for women), and in men SHBG decreased risk of erectile dysfunction (GCP = 0.42, p = 0.00032). Finally, we observed causality between SHBG and age at menopause ( $\beta = -0.756$ , p = 0.0010), suggesting SHBG levels modify reproductive phenotypes across female lifespan.

Overall, we detected relatively little robust statistical evidence for T levels being causal to most traits (Supplementary Data 12), though we noted also some nominally significant associations, like male free T reducing body fat (GCP = 0.50, p = 0.036). Yet, emphasizing the intricate relationship between hormone levels and metabolism<sup>69</sup>, we for instance detected obesity and liver function related traits being causal for T in females: triglycerides increasing free T levels, (GCP = -0.80, p = 1.4e-06) and GGT total T levels (GCP-0.60, p = 4.5e-06). In combination with the FinnGen data, these results thus further suggest that in most instances the connection between T and SHBG levels and complex traits and diseases is not straightforward.

#### Discussion

Since its discovery in the early 20th century, testosterone (T) has been proposed to modify phenotypes and diseases that differ between the sexes, due to the extensive male-female differences in circulating T levels. Besides the biological impacts of normal variation in T levels, the use of T replacement therapy in medical practice has been globally on rise, and the potential risks and benefits associated with T supplementation remain debated<sup>7</sup>. Although highly valuable, clinical trials to investigate the relationship of T levels to many complex phenotypes would be often infeasible or unethical to conduct. Here, using genetic data is an alternative means to estimate how population variability in baseline T levels connects with human health, leading to causal insights beyond the reported epidemiological relationships. In this study we combined data from over 625,000 participants of the FinnGen and UK Biobank cohorts to provide a broad and systematic perspective into the function of long term T exposure, a heritable trait in both sexes that can be proxied by using genetic data, as a regulator of health and disease in men and women. Taken together, supporting recent recommendations, our data suggests that the risks and benefits of using T as a medical



**Fig. 4 Results from the cross-sex PGS analyses. a**, **b** Show HR point estimates with 95% confidence intervals. **a** Cross-sex PGS associations with hypothyroidism, statin use, T2D and stroke. p < 0.05 suggests that the effects of a given PGS vary depending on sex. Green = total T PGS, blue = free T PGS. Dark green = HR in males, pink = HR in females. **b** Illustration of cases with statistical evidence for male and female PGSs having different effects (hirsutism, polycystic ovary syndrome (PCOS), infertility and post-menopausal bleeding (PMB). No evidence for such difference for breast and prostate cancers. Red estimates = female free T PGS, blue = male free T PGS. The data are based on 94,478 men and 122,986 women from FinnGen R5. Case and control numbers for each endpoint are available from Supplementary Data 9. \*Chi-Squared p < 0.05, \*\*Chi-Squared p < 0.0014, n.s. = not significant.



**Fig. 5 Genetic correlation and causal relationships between the 44 traits and phenotypes from public GWAS data based on LDSC and MR analyses.** The asterisks refer to significant genetic correlation between T/SHBG/free T and the studied phenotypes (\*p < 0.05 and \*\*p < 0.0011, after Bonferroni correction for 44 independent tests). Dark green symbol = males, pink = females. Red shades in boxes refer to positive genetic correlation (the same genetic factors increase both traits), and blue shades to negative genetic correlation (genetic factors increasing one trait decrease the other). Genetic correlation does not necessarily imply causality. Black squares mark significant causal relationships between the tested traits based on LCV and MR Egger analyses. Data is based on GWAS summary statistics from studies ranging from 18,759 MDD cases to over a million participants in risk tolerance GWASs, most studies including several hundreds of thousands of subjects (Supplementary Data 11). Green = total testosterone, white lilac = SHBG, blue = free testosterone. AAM age at menarche, ALAT alanine transaminase, ADHD attention deficit and hyperactivity disorder, CHD coronary heart disease, GGT gamma-glutamyl transferase, HBMD heel bone mineral density, MDD major depressive disorder, MPB male pattern baldness, RA rheumatoid arthritis, RHGS relative handgrip strength, WHR waist-to-hip ratio.

treatment should be carefully weighted, given T's complex and indirect relationship to most phenotypes and potential adverse and beneficial outcomes in both sexes<sup>7,26</sup>.

We observed three major themes regarding genetically determined T's contribution to disease. First, we detected causality between higher T levels and several sex-biased phenotypes with biological links to T, but less contribution to most other phenotypes, echoing experimental data and findings from recent MR studies<sup>23,26</sup>. Whereas our analyses supported causality of genetically determined T, e.g., to hormone-sensitive cancers and hirsutism, this did not apply for traits such as obesity, T2D or hypothyroidism. Accordingly, we stress the complex relationship of T levels with many metabolic and endocrine traits. The results echo the epidemiological observations that lower T levels correlate with increased risk of metabolic disease in men, while in women the opposite hold true<sup>70,71</sup>. However, our data shows that these associations to metabolic health may reflect widespread genetic pleiotropy, implying complex etiology as opposed to T action. We further estimate that normal heritable variation in T levels—contrary to popular beliefs—has only modest effects on many phenotypes. In the light of our study, despite the potential behavioral impact of hormonal processes, for example the concept of explaining the risk for many temperamental and neurological phenotypes with heritable differences in adult T levels<sup>72,73</sup> appears unsubstantiated.

Secondly, we highlight how our data reflects the correlations between total T, SHBG and free T levels in both sexes. Total T in men correlates with SHBG levels epidemiologically<sup>13–15</sup> and at the genetic level, meaning that same genetic variants affect both traits. Underscoring the contribution of SHBG to total T associations, we found SHBG causal for total T levels in males. Especially for many metabolic traits, SHBG-either directly or through its metabolic network-appeared to modify some of total T's associations and causality estimates. Consistent with potentially divergent biological effects for the SHBG bound and unbound T, we however observed distinct association profiles for total and free T fractions in men. For women, the situation was the opposite: whereas total T associations seemed largely independent of SHBG action, for many metabolic traits the high dependency of calculated free T on SHBG levels was evident.

Finally, we report that genetic differences affecting T levels seemed to affect disease risk especially in women, including PCOS-related endpoints like hirsutism, PMB and infertility. Although our study suggests that genetically determined T and PCOS may be partly different entities, we unfortunately cannot state if genetically determined T has fully PCOS-independent effects on these traits, since our study may be limited by PCOS being potentially underdiagnosed in FinnGen, or the diagnosed PCOS cases representing only a subpopulation of this heterogenous disorder<sup>74,75</sup>. In addition, all these traits correlate with obesity<sup>32</sup>, that in turn correlates with PCOS and T levels (Supplementary Fig. 10 and Supplementary Data 8), and we may have not been able to fully account for such confounding.

Out of the metabolic traits, our T2D results that are in line with clinical trials but contrast recent genetic findings<sup>23,33,76</sup> merit an extended discussion. We note that some of our MR analyses also supported a causal link between T and T2D, like previously shown<sup>23,24</sup>. However, the MR models developed to address confounding by genetic pleiotropy, including variants effects on SHBG, did not support causality. Finally, the crosssex analyses, relying on the T PGSs not predicting T levels in the opposite sex, strongly suggested that the T/T2D associations do not reflect causality. For both male and female specific PGSs, the T/T2D associations remained similar also in the other sex, meaning that the original sex-specific associations unlikely stem from T action. Since we detected limited evidence also for SHBG's causality to T2D, echoing some recent findings<sup>77</sup>, our data thus suggests that overall genetic pleiotropy and factors affecting, e.g., BMI or liver function, may be more likely accountable for the connection between T levels and T2D.

Having comprehensively mapped the impacts of T across diverse complex disease and traits, we continued to speculate on the role of T levels as a contributor to the male-female differences. Overall, higher T levels associated with typical male characteristics. For instance, our work suggested T is causally connected to higher hemoglobin and bone strength and lower body fat, backed up by previous experimental observations<sup>1,78,79</sup>. Extrapolating from the MR results, we estimate that ~10-20% of the mean difference in hemoglobin levels between men and women may result from average differences in free T levels, consistent with the notion that T directly affects male-female differences in, e.g., athletic capacity<sup>1</sup>. Moreover, higher T levels were found causal to masculine external features like hirsutism and baldness. The fact that T levels only partly explain these traits and the limited evidence for the causal involvement of T for many traits showing sex differences like cardiovascular disease however suggests that sex differences in disease prevalence are not attributable to a linear relationship between T levels and a phenotype. Considering the 7-15× differences in T levels between an average male and female, that should lead to vast differences in disease incidence between the sexes if adult T levels were a major determinant for health, this is only to be expected.

We however provide some unique insight into the genetic causes behind the sex differences in T levels. Cross-sex genetic correlations for traits related to fitness (e.g., reproductive success) are generally expected to be low, due to potentially conflicting evolutionary pressures<sup>80</sup>. We indeed associated T positively with number of offspring in men, but negatively with reproductive health in both pre- and postmenopausal women. Besides the wellestablished biological dichotomy between the sexes, leading to vast differences in T levels, we may speculate that there exists a selective advantage in relation to reproductive success to maintain adequate T levels in males, whereas the opposite may be true for females. This in turn might partly promote the widespread sex differences across multiple traits, and the unique sex-specificity of genetics of T. In the cross-sex PGS analyses of disease endpoints common to both sexes, we however found only one case with clear evidence of antagonistic associations for a T PGS. The malespecific total T PGS protected from stroke in men, whereas the same PGS had opposite effects on stroke risk in women. Although we cannot conclude that the action of T drives these associations, the result agrees with sex-specificity in the genetic disease mechanisms for stroke<sup>66</sup>.

Although based on extensive data sets, our study has limitations. The UK Biobank data is based immunoassays, which may have limited use in clinical settings. Yet we show that the T measures from two distinct visits are highly correlated in both sexes, which together with the robust genetic findings illustrates that the data allows for estimating how variable T levels relate to adult health. We however stress that our genetic data clearly suggests that many different biological processes are involved in determining normal variation in T levels, serum T levels likely serving only as a proxy for T action in tissues. Furthermore, although the PGS for total T and SHBG, (based on participants aged between 39-72 years, women mostly post-menopausal) explained a substantial fraction of the trait variance also in younger individuals ( $R^2$  9.2% for SHBG in males in the YFS), for example male free T PGS seemed to yield less accurate predictions (1.1% in YFS). Since the majority of the variance remains uncaptured by the PGSs, we cannot draw conclusions about the function of T based on the PGSs alone, and results may not be fully generalizable to other than European populations. Nonetheless, we seem to have been well powered to detect any substantial effects for the PGS (Methods and Supplementary Fig. 7). In line with the data from other complex traits, it thus seems that for most persons information about T PGSs may have limited use, but for the extremes of distribution such information may prove clinically valuable in the future<sup>81</sup>.

Given the combination of analysis strategies—for the MR estimates, 1 SD change in circulating T levels as the basic unit—

we were well positioned to detect any substantial effects normal variation in adult T levels, excluding clinical conditions, would have on disease and health. Yet we remind that the MR methods have some important limitations, including susceptibility to genetic pleiotropy (Methods), and these did not always agree upon causality. We further note that the emphasis of our study has been on adult T levels, whereas some of the effects of T might be organizational, acting during specific developmental windows. Our setting thus may not allow for assessing the effects of fetal T exposure, which may be critical, e.g., for neurological traits<sup>82</sup>, or serum T may not be the optimal androgen to assess such risk associations. We also emphasize that our results are based on normal, heritable variation in T levels, not, e.g., on supraphysiological synthetic T use wrecking baseline hormonal balance, which, combined with specific external stimuli such as exercise regime are linked with more drastic effects, e.g., on muscle strength and health<sup>83</sup>. Additionally, many of T's effects depend on its conversion to estradiol also in males, and we cannot rule this out as a potential confounder in our study. Finally, the data used in our study does not allow for assessing the effects of acute changes in hormone secretion, including those happening during different phases of the menstrual cycle in females or upon certain environmental stimuli, and personal differences in responses to such fluctuations.

Despite these challenges, in this study we highlight several previously unreported, albeit often expected relationships with genetically determined T levels, human health, and sex differences. Besides the gained medical insight, underscoring some critical factors that should be considered when assessing these connections, our study therefore provides a novel reference point for future genetic and epidemiological studies on T.

#### Data availability

The source data for figures is available in Supplementary Data: for Fig. 1— Supplementary Data 2; Fig. 2—Supplementary Data 7; Fig. 3—Supplementary Data 8; Fig. 4—Supplementary Data 9; Fig. 5—Supplementary Data 10 and 11. Full genetic and clinical data from FinnGen and the UK Biobank are available for researchers by application, https://www.finngen.fi/en/access\_results and https://www.ukbiobank.ac.uk/. For YFS, see https://youngfinnsstudy.utu.fi/index.html. GWAS summary statistics for total T, SHBG, FAI and free T based on the UK Biobank data will be available from the GWAS catalog (https://www.ebi.ac.uk/gwas) with submission ID GCP000476. Data from publicly available GWASs can be downloaded from the source repositories with references listed in Supplementary Data 11.

#### Code availability

The full genotyping and imputation protocol for FinnGen is described at https://doi.org/ 10.17504/protocols.io.nmndc5e<sup>84</sup>.

Received: 17 September 2021; Accepted: 7 December 2022; Published online: 18 January 2023

#### References

- Handelsman, D. J., Hirschberg, A. L. & Bermon, S. Circulating testosterone as the hormonal basis of sex differences in athletic performance. *Endocr. Rev* 39, 803–829 (2018).
- Tyagi, V., Scordo, M., Yoon, R. S., Liporace, F. A. & Greene, L. W. Revisiting the role of testosterone: are we missing something? *Rev. Urol.* 19, 16–24 (2017).
- Davis, S. R. & Wahlin-Jacobsen, S. Testosterone in women—the clinical significance. *Lancet Diabetes Endocrinol.* 3, 980–992 (2015).
- Vandenput, L. & Ohlsson, C. Genome-wide association studies on serum sex steroid levels. *Mol. Cell. Endocrinol.* 382, 758–766 (2014).
- Kloner, R. A., Carson, C., Dobs, A., Kopecky, S. & Mohler, E. R. Testosterone and cardiovascular disease. J. Am. Coll. Cardiol 67, 545–557 (2016).

- Gagliano-Jucá, T. & Basaria, S. Testosterone replacement therapy and cardiovascular risk. *Nat. Rev. Cardiol.* 16, 555–574 (2019).
- Bhasin, S. et al. Testosterone therapy in men with hypogonadism: an Endocrine Society Clinical Practice Guideline. J. Clin. Endocrinol. Metab. 103, 1715–1744 (2018).
- Celec, P., Ostatníková, D. & Hodosy, J. On the effects of testosterone on brain behavioral functions. *Front. Neurosci.* 9, 12–12 (2015).
- Venturelli, E. et al. Observational study on the prognostic value of testosterone and adiposity in postmenopausal estrogen receptor positive breast cancer patients. *BMC Cancer* 18, 651–651 (2018).
- Stattin, P. et al. High levels of circulating testosterone are not associated with increased prostate cancer risk: a pooled prospective study. *Int. J. Cancer* 108, 418–424 (2004).
- Eisenegger, C., Naef, M., Snozzi, R., Heinrichs, M. & Fehr, E. Prejudice and truth about the effect of testosterone on human bargaining behaviour. *Nature* 463, 356–359 (2010).
- Clark, R. V. et al. Large divergence in testosterone concentrations between men and women: Frame of reference for elite athletes in sex-specific competition in sports, a narrative review. *Clin. Endocrinol.* **90**, 15–22 (2019).
- 13. Goldman, A. L. et al. A reappraisal of testosterone's binding in circulation: physiological and clinical implications. *Endocr. Rev* 38, 302–324 (2017).
- Cunningham, S. K., Loughlin, T., Culliton, M. & McKenna, T. J. The relationship between sex steroids and sex-hormone-binding globulin in plasma in physiological and pathological conditions. *Ann. Clin. Biochem.* 22, 489–497 (1985).
- Mendel, C. M. The free hormone hypothesis: a physiologically based mathematical model. *Endocr. Rev.* 10, 232–274 (1989).
- Rosner, W., Auchus, R. J., Azziz, R., Sluss, P. M. & Raff, H. Utility, limitations, and pitfalls in measuring testosterone: an Endocrine Society Position Statement. J. Clin. Endocrinol. Metab. 92, 405–413 (2007).
- Pasquali, R. The hypothalamic-pituitary-adrenal axis and sex hormones in chronic stress and obesity: pathophysiological and clinical aspects. *Ann. NY Acad. Sci.* 1264, 20–35 (2012).
- Trost, L. W. & Mulhall, J. P. Challenges in testosterone measurement, data interpretation, and methodological appraisal of interventional trials. *J. Sex. Med.* 13, 1029–1046 (2016).
- Ohlsson, C. et al. Genetic determinants of serum testosterone concentrations in men. PLoS Genet. 7, e1002313 (2011).
- Harris, J. A., Vernon, P. A. & Boomsma, D. I. J. B. G. The heritability of testosterone: a study of dutch adolescent twins and their parents. *Behav. Genet.* 28, 165–171 (1998).
- Ruth, K. S. et al. Genome-wide association study with 1000 genomes imputation identifies signals for nine sex hormone-related phenotypes. *Eur. J. Hum. Genet.* 24, 284–290 (2016).
- Koenis, M. M. G. et al. Longitudinal study of hormonal and physical development in young twins. J. Clin. Endocrinol. Metab. 98, E518–E527 (2013).
- 23. Ruth, K. S. et al. Using human genetics to understand the disease impacts of testosterone in men and women. *Nat. Med.* **26**, 252–258 (2020).
- 24. Flynn, E. et al. Sex-specific genetic effects across biomarkers. *Eur. J. Hum. Genet.* **29**, 154–163 (2020).
- Sinnott-Armstrong, N., Naqvi, S., Rivas, M. & Pritchard, J. K. GWAS of three molecular traits highlights core genes and pathways alongside a highly polygenic background. *eLife* 10, e58615 (2021).
- Mohammadi-Shemirani, P. et al. Effects of lifelong testosterone exposure on health and disease using Mendelian randomization. *eLife* 9, e58914 (2020).
- Hayes, B.L. et al. Do sex hormones confound or mediate the effect of chronotype on breast and prostate cancer? A Mendelian randomization study. *PLoS Genet.* 18, e1009887 (2021).
- Harrison, S., Davies, N. M., Howe, L. D. & Hughes, A. Testosterone and socioeconomic position: Mendelian randomization in 306,248 men and women participants of UK Biobank. *Sci. Adv.* 7, eabf8257 (2021).
- 29. Bycroft, C. et al. The UK Biobank resource with deep phenotyping and genomic data. *Nature* **562**, 203–209 (2018).
- Vermeulen, A., Verdonck, L. & Kaufman, J. M. A critical evaluation of simple methods for the estimation of free testosterone in serum. *J. Clin. Endocrinol. Metab.* 84, 3666–3672 (1999).
- Emadi-Konjin, P., Bain, J. & Bromberg, I. L. Evaluation of an algorithm for calculation of serum "Bioavailable" testosterone (BAT). *Clin. Biochem.* 36, 591–596 (2003).
- 32. Venkatesh, S. S. et al. Obesity and risk of female reproductive conditions: a Mendelian randomisation study. *PLoS Med.* **19**, e1003679 (2022).
- 33. Grossmann, M., Hoermann, R., Wittert, G. & Yeap, B. B. Effects of testosterone treatment on glucose metabolism and symptoms in men with type 2 diabetes and the metabolic syndrome: a systematic review and metaanalysis of randomized controlled clinical trials. *Clin. Endocrinol.* 83, 344–351 (2015).

#### ARTICLE

- 34. Sinnott-Armstrong, N. et al. Genetics of 35 blood and urine biomarkers in the UK Biobank. *Nat. Genet.* **53**, 185–194 (2021).
- Watanabe, K., Taskesen, E., van Bochoven, A. & Posthuma, D. Functional mapping and annotation of genetic associations with FUMA. *Nat. Commun.* 8, 1826 (2017).
- Kanehisa, M., Sato, Y., Kawashima, M., Furumichi, M. & Tanabe, M. KEGG as a reference resource for gene and protein annotation. *Nucleic Acids Res.* 44, D457–D462 (2015).
- 37. Pickrell, J. K. et al. Detection and interpretation of shared genetic influences on 42 human traits. *Nat. Genet.* **48**, 709–717 (2016).
- Benner, C. et al. Prospects of fine-mapping trait-associated genomic regions by using summary statistics from genome-wide association studies. *Am. J. Hum. Genet.* 101, 539–551 (2017).
- Vilhjálmsson, B. J. et al. Modeling linkage disequilibrium increases accuracy of polygenic risk scores. Am. J. Hum. Genet. 97, 576–592 (2015).
- Bulik-Sullivan, B. et al. An atlas of genetic correlations across human diseases and traits. Nat. Genet. 47, 1236–1241 (2015).
- Mars, N. J. et al. Polygenic and clinical risk scores and their impact on age at onset of cardiometabolic diseases and common cancers. *Nat. Med.* 26, 549–557 (2020).
- Vihma, V. et al. Metabolism of sex steroids is influenced by acquired adiposity —a study of young adult male monozygotic twin pairs. J. Steroid Biochem. Mol. Biol. 172, 98-105 (2017).
- Eriksson, J. et al. Causal relationship between obesity and serum testosterone status in men: a bi-directional mendelian randomization analysis. *PLoS ONE* 12, e0176277 (2017).
- Burger, H. G., Hale, G. E., Robertson, D. M. & Dennerstein, L. A review of hormonal changes during the menopausal transition: focus on findings from the Melbourne Women's Midlife Health Project. *Hum. Reprod. Updat.* 13, 559–565 (2007).
- 45. Ruth, K. S. et al. Genetic insights into biological mechanisms governing human ovarian ageing. *Nature* **596**, 393–397 (2021).
- Howe, L. J. et al. Within-sibship genome-wide association analyses decrease bias in estimates of direct genetic effects. *Nat. Genet.* 54, 581–592 (2022).
- Richardson, T. G., Harrison, S., Hemani, G. & Davey Smith, G. An atlas of polygenic risk score associations to highlight putative causal relationships across the human phenome. *eLife* 8, e43657 (2019).
- Leinonen, J. T. et al. LIN28B affects gene expression at the hypothalamicpituitary axis and serum testosterone levels. *Sci. Rep.* 9, 18060 (2019).
- Sivakumaran, S. et al. Abundant pleiotropy in human complex diseases and traits. Am. J. Hum. Genet. 89, 607–618 (2011).
- Chesmore, K., Bartlett, J. & Williams, S. M. The ubiquity of pleiotropy in human disease. *Hum. Genet.* 137, 39–44 (2018).
- 51. Fawcett, K.A. et al. Pleiotropic associations of heterozygosity for the
- SERPINA1 Z allele in the UK Biobank. *ERJ Open Res.* 7, 00049-02021 (2021).
  Oetjens, M. T. et al. Evidence for extensive pleiotropy among pharmacogenes. *Pharmacogenomics* 17, 853–866 (2016).
- O'Connor, L. J. & Price, A. L. Distinguishing genetic correlation from causation across 52 diseases and complex traits. *Nat. Genet.* **50**, 1728–1734 (2018).
- Burgess, S. & Thompson, S. G. Interpreting findings from Mendelian randomization using the MR-Egger method. *Eur. J. Epidemiol.* 32, 377–389 (2017).
- Morrison, J., Knoblauch, N., Marcus, J. H., Stephens, M. & He, X. Mendelian randomization accounting for correlated and uncorrelated pleiotropic effects using genome-wide summary statistics. *Nat. Genet.* 52, 740–747 (2020).
- Rees, J. M. B., Wood, A. M. & Burgess, S. Extending the MR-Egger method for multivariable Mendelian randomization to correct for both measured and unmeasured pleiotropy. *Stat. Med.* 36, 4705–4718 (2017).
- Davey Smith, G. & Hemani, G. Mendelian randomization: genetic anchors for causal inference in epidemiological studies. *Hum. Mol. Genet.* 23, R89–R98 (2014).
- Astle, W. J. et al. The allelic landscape of human blood cell trait variation and links to common complex disease. *Cell* 167, 1415.e19–1429.e19 (2016).
- Hemani, G. et al. The MR-Base platform supports systematic causal inference across the human phenome. *eLife* 7, e34408 (2018).
- Yavorska, O. O. & Burgess, S. MendelianRandomization: an R package for performing Mendelian randomization analyses using summarized data. *Int. J. Epidemiol* 46, 1734–1739 (2017).
- Taieb, J. et al. Testosterone measured by 10 immunoassays and by isotopedilution gas chromatography-mass spectrometry in sera from 116 men, women, and children. *Clin. Chem.* 49, 1381–1395 (2003).
- Legro, R. S. et al. Total testosterone assays in women with polycystic ovary syndrome: precision and correlation with hirsutism. *J. Clin. Endocrinol. Metab.* 95, 5305–5313 (2010).
- Martin, K. A. et al. Evaluation and treatment of hirsutism in premenopausal women: an Endocrine Society Clinical Practice Guideline. J. Clin. Endocrinol. Metab. 103, 1233–1257 (2018).

- Clarke, M. A. et al. Association of endometrial cancer risk with postmenopausal bleeding in women: a systematic review and meta-analysis. *JAMA Intern. Med.* 178, 1210–1222 (2018).
- Tyrmi, J. S. et al. Leveraging Northern European population history: novel low-frequency variants for polycystic ovary syndrome. *Hum. Reprod.* 37, 352–365 (2022).
- 66. Traylor, M. et al. Differences in common genetic predisposition to ischemic stroke by age and sex. *Stroke* **46**, 3042–3047 (2015).
- van Rheenen, W., Peyrot, W. J., Schork, A. J., Lee, S. H. & Wray, N. R. Genetic correlations of polygenic disease traits: from theory to practice. *Nat. Rev. Genet.* 20, 567–581 (2019).
- Choi, M. H., Yoo, Y. S. & Chung, B. C. Biochemical roles of testosterone and epitestosterone to 5α-reductase as indicators of male-pattern baldness. *J. Invest. Dermatol.* 116, 57–61 (2001).
- Palmisano, B. T., Zhu, L., Eckel, R. H. & Stafford, J. M. Sex differences in lipid and lipoprotein metabolism. *Mol. Metab.* 15, 45–55 (2018).
- Grossmann, M. Low testosterone in men with type 2 diabetes: significance and treatment. J. Clin. Endocrinol. Metab. 96, 2341–2353 (2011).
- Janssen, I., Powell, L. H., Crawford, S., Lasley, B. & Sutton-Tyrrell, K. Menopause and the metabolic syndrome: the study of women's health across the nation. Arch. Intern. Med. 168, 1568–1575 (2008).
- McHenry, J., Carrier, N., Hull, E. & Kabbaj, M. Sex differences in anxiety and depression: role of testosterone. *Front. Neuroendocrinol.* 35, 42–57 (2014).
- O'Connor, D. B., Archer, J. & Wu, F. C. W. Effects of testosterone on mood, aggression, and sexual behavior in young men: a double-blind, placebocontrolled, cross-over study. J. Clin. Endocrinol. Metab. 89, 2837–2845 (2004).
- Actkins, K. E. V. et al. Characterizing the clinical and genetic spectrum of polycystic ovary syndrome in electronic health records. J. Clin. Endocrinol. Metab. 106, 153–167 (2020).
- Dapas, M. et al. Distinct subtypes of polycystic ovary syndrome with novel genetic associations: an unsupervised, phenotypic clustering analysis. *PLoS Med* 17, e1003132 (2020).
- Huang, G. et al. Long-term testosterone administration on insulin sensitivity in older men with low or low-normal testosterone levels. J. Clin. Endocrinol. Metab. 103, 1678–1685 (2018).
- 77. Wang, Q. et al. Sex hormone-binding globulin associations with circulating lipids and metabolites and the risk for type 2 diabetes: observational and causal effect estimates. *Int. J. Epidemiol.* **44**, 623–637 (2015).
- Snyder, P. J. et al. Effect of testosterone treatment on volumetric bone density and strength in older men with low testosterone: a controlled clinical trial. *JAMA Intern. Med.* 177, 471–479 (2017).
- 79. Roy, C. N. et al. Association of testosterone levels with anemia in older men: a controlled clinical trial. *JAMA Intern. Med.* **177**, 480–490 (2017).
- Connallon, T. & Matthews, G. Cross-sex genetic correlations for fitness and fitness components: connecting theoretical predictions to empirical patterns. *Evol. Lett.* 3, 254–262 (2019).
- 81. Lewis, C. M. & Vassos, E. Prospects for using risk scores in polygenic medicine. *Genome Med.* 9, 96 (2017).
- Suderman, M. et al. Sex-associated autosomal DNA methylation differences are wide-spread and stable throughout childhood. Preprint at *bioRxiv* https:// doi.org/10.1101/118265 (2017).
- Bhasin, S. et al. The effects of supraphysiologic doses of testosterone on muscle size and strength in normal men. N. Engl. J. Med. 335, 1–7 (1996).
- 84. Pärn, K. et al. Genotype imputation workflow v3.0 V.1. (Protocols.io, 2018).

#### Acknowledgements

The research has been conducted using the UK Biobank Resource under application number 22627. The FinnGen project is funded by two grants from Business Finland (HUS 4685/31/2016 and UH 4386/31/2016) and eleven industry partners (AbbVie Inc, AstraZeneca UK Ltd, Biogen MA Inc, Celgene Corporation, Celgene International II Sarl, Genentech Inc, Merck Sharp & Dohme Corp, Pfizer Inc., GlaxoSmithKline, Sanofi, Maze Therapeutics Inc., Janssen Biotech Inc). The funders had no role in study design, data collection and analysis, decision to publish or preparation of the manuscript. The following biobanks are acknowledged for collecting the FinnGen project samples: Auria Biobank (https://www.auria.fi/biopankki), THL Biobank (https://thl.fi/fi/web/thlbiopankki), Helsinki Biobank (https://www.terveyskyla.fi/helsinginbiopankki), Biobank Borealis of Northern Finland (https://www.oulu.fi/university/node/38474), Finnish Clinical Biobank Tampere (https://www.tays.fi/en-US/Research\_and\_development/ Finnish\_Clinical\_Biobank\_Tampere), Biobank of Eastern Finland (https://itasuomenbiopankki.fi), Central Finland Biobank (https://www.ksshp.fi/fi-FI/Potilaalle/ Biopankki), Finnish Red Cross Blood Service Biobank (https://www.veripalvelu.fi/ verenluovutus/biopankkitoiminta) and Terveystalo Biobank (https://www.terveystalo. com/fi/Yritystietoa/Terveystalo-Biopankki/Biopankki/). All Finnish Biobanks are members of BBMRI.fi infrastructure (www.bbmri.fi). The Young Finns Study has been financially supported by the Academy of Finland: grants 322098 (T.L), 286284, 134309 (Eye), 126925, 121584, 124282, 129378 (Salve), 117787 (Gendi), and 41071 (Skidi); the Social Insurance Institution of Finland; Competitive State Research Financing of the

Expert Responsibility area of Kuopio, Tampere and Turku University Hospitals (grant X51001); Juho Vainio Foundation; Paavo Nurmi Foundation; Finnish Foundation for Cardiovascular Research; Finnish Cultural Foundation; The Sigrid Juselius Foundation; Tampere Tuberculosis Foundation; Emil Aaltonen Foundation; Yrjo Jahnsson Foundation; Signe and Ane Gyllenberg Foundation; Diabetes Research Foundation of Finnish Diabetes Association; This project has received funding from the European Union's Horizon 2020 research and innovation program under grant agreements No 848146 for To Aition and grant agreement 755320 for TAXINOMISIS; European Research Council (grant 742927 for MULTIEPIGEN project); Tampere University Hospital Supporting Foundation and Finnish Society of Clinical Chemistry. We greatly thank all UK Biobank, the Young Finns study, and FinnGen participants, as well as the principal investigators, laboratory personnel and data management teams behind these efforts. This work was supported by the Academy of Finland (https://www.aka.fi/en/) grants 331671 to N.M., 312072 to T. Tuomi, 288509 and 319181 to M.P., and 315589 and 320129 to T. Tukiainen, and Academy of Finland Center of Excellence in Complex Disease Genetics grants 312075 to M.D., 312062 and 336820 to S.Ripatti and 312076 to M.P.), by the Finnish Foundation for Cardiovascular Research (https://www.sydantutkimussaatio.fi/en) (S.Ripatti), the Sigrid Juselius Foundation (https://sigridjuselius.fi/en/) (S.Ripatti, M.P., T. Tukiainen) and University of Helsinki (https://www.helsinki.fi/en) HiLIFE Fellow and Grand Challenge grants (S.R. and M.P.), and 3-year research project grant (T. Tukiainen).

#### Author contributions

J.T.L. and T. Tukiainen conceptualized the study, designed the analysis plan and wrote the manuscript. N.M. calculated the PGSs for FinnGen and performed PGS associations. L.E.L. performed GWASs in the UK Biobank. A.A.-O. calculated PGSs and assisted in the analysis of YFS. S. Ruotsalainen performed GWAS in FinnGen, if desired summary statistics were not directly available. J.T.L. and T. Tukiainen performed the rest of the analyses. T.L., M.K., O.R. and FinnGen consortium provided data, T.P. and T. Tuomi clinical and M.P. statistical expertise. N.M., T.P., T. Tuomi, M.D., S. Ripatti and M.P. provided insightful comments on the study and text improving the manuscript. All authors participated in writing and reviewing the manuscript, and approved the final version of the manuscript.

#### **Competing interests**

The authors declare no competing interests.

#### **Additional information**

**Supplementary information** The online version contains supplementary material available at https://doi.org/10.1038/s43856-022-00226-0.

Correspondence and requests for materials should be addressed to Taru Tukiainen.

Peer review information Communications Medicine thanks the anonymous reviewers for their contribution to the peer review of this work. Peer reviewer reports are available.

Reprints and permission information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2023

#### **FinnGen Consortium**

Nina Mars<sup>17</sup>, Sanni Ruotsalainen<sup>17</sup>, Mika Kähönen<sup>5,18</sup>, Terhi Piltonen<sup>19</sup>, Tiinamaija Tuomi<sup>17,20,21,22,23</sup>, Mark Daly<sup>17,24</sup>, Samuli Ripatti<sup>17,24,25</sup> & Taru Tukiainen<sup>17</sup>

<sup>17</sup>Institute for Molecular Medicine Finland (FIMM), HiLIFE, University of Helsinki, Helsinki, Finland. <sup>18</sup>Faculty of Medicine and Health Technology, Tampere University, Tampere, Finland. <sup>19</sup>Department of Obstetrics and Gynecology, PEDEGO Research Unit, Medical Research Centre, Oulu University Hospital, University of Oulu, Oulu, Finland. <sup>20</sup>Department of Endocrinology, Abdominal Centre, Helsinki University Hospital, Helsinki, Finland. <sup>21</sup>Folkhalsan Research Center, Helsinki, Finland. <sup>22</sup>Research Programs Unit, Clinical and Molecular Metabolism, University of Helsinki, Helsinki, Finland. <sup>23</sup>Lund University Diabetes Centre, Department of Clinical Sciences Malmö, Lund University, Malmö, Sweden. <sup>24</sup>Broad Institute of MIT and Harvard, Cambridge, MA, USA. <sup>25</sup>Department of Public Health, Faculty of Medicine, University of Helsinki, Finland. A full list of members and their affiliations appears in the Supplementary Information.